# Comprehensive Evaluation of Multisource Soil Moisture Products in a Managed Agricultural Region: An Integrated Hydrologic Modeling Approach

Zheng Lu<sup>®</sup>, Yuan He, Shuyan Peng, and Xiaofan Yang<sup>®</sup>

Abstract-Soil moisture (SM) is important in understanding Earth's hydrologic cycles. However, an overall performance of multisource SM products is still unclear due to a lack of comprehensive validation. Using SM simulated by hydrologic models as a reference to perform validation is a promising alternative since SM simulation is not restricted by its coverage or scale. In this study, an integrated hydrologic model (ParFlow-CLM) forced by hydrometeorological and agricultural water use reanalysis data is built in the middle reaches of the Heihe River Basin (HRB), a typical managed agricultural region in Northwestern China. Using the ParFlow-CLM simulated SM data as the validating reference, ten SM products, including four single-source RS SM, three merged SM, and three assimilated SM products are systematically assessed by a comprehensive evaluation framework composed of fifteen statistical performance indicators. For validation, the results show that ParFlow-CLM SM simulations agree with the time domain reflectometry probe and cosmic ray neutron probes observations. The merged SM and assimilated SM outperform the single-source RS SM products if compared with ParFlow-CLM simulated data. Results from the intercomparison demonstrate that hydraulic conductivity and leaf area index are the dominant factors in SM spatial variations based on the generalized additive model. The statistical linkage indicates that mean absolute deviation, uncertainty at 95% (U95), Nash-Sutcliffe's efficiency, and combined performance index serve as substitutes for quantifying the relative uncertainty of SM. This study paves a way for model-data intercomparison of SM products in the HRB as well as in other arid and semiarid basins.

*Index Terms*—Heihe River Basin (HRB), integrated hydrologic model, intercomparison, remote sensing (RS) products, soil moisture (SM), validation.

The authors are with the State Key Laboratory of Earth Surface Processes and Resource Ecology, School of Natural Resources, Faculty of Geographical Science, Beijing Normal University, Beijing 100875, China (e-mail: legend.lz@mail.bnu.edu.cn; yuanhe@mail.bnu.edu.cn; sypeng@mail.bnu.edu. cn; xfyang@bnu.edu.cn).

Digital Object Identifier 10.1109/JSTARS.2023.3289649

#### I. INTRODUCTION

**S** OIL moisture (SM) is a key element linking the atmosphere and terrestrial ecosystems [1]. Comprehensive validation and intercomparison of SM products could help improve many hydrometeorological applications, such as precipitation forecasting [2], flood and drought monitoring [3], and climate change studies [4]. Currently, SM estimates are primarily obtained by in situ measurements, satellite retrievals, data assimilations, and hydrologic model simulations.

In recent decades, microwave remote sensing (RS), especially passive microwave approaches, has become an effective technology for SM global mapping owing to its high sensitivity to surface SM [5], [6], [7]. The passive microwave SM products include advanced microwave scanning radiometer (AMSR-E) on the Earth observing system (EOS) Aqua satellite [8], subsequent advanced microwave scanning radiometer 2 (AMSR2) onboard the global change observation mission (GCOM-W1) spacecraft [9], and microwave imaging radiometer with aperture synthesis onboard the soil moisture and ocean salinity (SMOS) satellite [10]. These satellite SM products, which have global coverage at a certain revisit time step, are feasible in global-scale applications, as they offer a snapshot of the evolving land surface processes at a specific local time [11].

However, single RS-source SM suffers from many deficiencies in practice, such as discontinuity in time series and higher uncertainty applied in dense vegetation cover areas [12]. The European Space Agency (ESA) Climate Change Initiative (CCI) project considers an integration of active and/or passive SM products and provides three effective merged SM products, which aims to overcome the drawbacks of the single RS-source SM [13]. The CCI SM dataset consists of three SM products by different blending techniques [14].

Additionally, as a multidisciplinary approach that considers several land surface processes, SM products by data assimilation are an effective tool for providing spatiotemporally continuous multilayer SM at different scales. Two of the widely used assimilated SM products are NASA's Global Land Data Assimilation System (GLDAS) SM product [15] and the Global Land Evaporation Amsterdam Model (GLEAM) SM product [16]. For better understanding and utilizing the above-mentioned SM products, an effective validation of SM products is

This work is licensed under a Creative Commons Attribution 4.0 License. For more information, see https://creativecommons.org/licenses/by/4.0/

Manuscript received 8 May 2023; revised 31 May 2023; accepted 23 June 2023. Date of publication 26 June 2023; date of current version 12 July 2023. This work was supported in part by the Strategic Priority Research Program of the Chinese Academy of Sciences under Grant XDA20100104, in part by Second Tibetan Plateau Scientific Expedition and Research Program under Grant 2019QZKK0306, in part by the National Natural Science Foundation of China under Grant 41877183 and Grant 42077172, and in part by Dragon 5 Cooperation 2020–2024 under Grant 59343. (*Corresponding author: Xiaofan Yang.*)

commonly performed by comparisons against reference data, aiming at obtaining systematic and random errors or deviations in SM products and accessing quantitative statistics on the SM quality [17].

One key issue in validating the SM products is the reference data selection. The reference data, which are assumed to be reliable in representing the "validating referenced" SM value, are commonly provided from three sources: field campaigns, in situ networks, and model simulations [18], [19]. in situ measurements (e.g., time domain reflectometry probes or ground penetrating radar probes) that arguably provide the most precise SM estimates are usually limited in spatial expansion [20]. However, the mismatch of spatial-temporal scales in targeted RS SM data between field campaigns and in situ networks are inevitable [18], [21], [22]. Using SM simulated by hydrological or land-surface models as a reference to perform validation is a promising alternative since SM simulation is not restricted by its coverage or scale [18]. Model simulations can provide regional SM maps with spatial-temporal continuity [23], [24]. Another advantage of using model simulations is that the model can output several relevant hydrological variables which can be used to analyze the corresponding responses in SM variations, such as evapotranspiration (ET) and infiltration (I) [25]. Usually, precipitation (from Tropical Rainfall Measuring Mission and Global Precipitation Measurement precipitations datasets) is utilized as an auxiliary variable alone to help analyze water balance in SM validation [26]. However, clouds or atmospheric distortions usually cause the time series discontinuity of precipitation, and thus spatial-temporal distributions of I cannot be directly obtained [27]. Vertical water exchange between the atmosphere and the surface soil layer is equal to I-ET, which is actually the net downward flux [28]. Therefore, it is important to obtain the correlation between SM responses and I-ET variations in the same framework.

Another important issue in validating the SM products is the evaluation method and assessment indicators. According to the work in [26] and [29], methods in SM validation are generally classified into two categories: direct and indirect evaluations. Direct evaluation usually uses statistical performance indicators calculated by comparing reference data against targeted SM products. Indirect evaluation is mainly based on cross-validation and consistency analysis of spatiotemporal trends using impact factors [30], [31]. It is noted that both direct and indirect evaluation methodologies need to improve. For direct evaluation, the shortcoming is that most SM validation research usually uses limited statistical performance indicators to assess the comparison results. Gruber et al. [18] summarized the current direct validation methods for SM products. Gueymard [32] provided an important review of validation methodologies and statistical performance indicators in solar project fields. Both reviews not only provided a theoretical background about SM validation studies but also proactively and effectively formed a validation framework. It is then feasible to transform the methodology system from Gueymard [32] to SM product validation. For indirect evaluation, the weakness is neglecting the linkage between cross-validation methods and indices in the direct validation. Liu et al. [31] reported that the three-cornered hat (TCH) based

relative uncertainty (RU) and root mean square error (RMSE) indices have good consistencies. However, the correlations of RUs and director performance indicators as well as the uniformities in direct and indirect results are still unknown [33]. It is therefore important to obtain their mathematical relationships.

The novelties of the current study include: 1) providing a reference SM from an integrated hydrological model with superior accuracy and high spatial-temporal resolution over a modern agricultural area, 2) developing a validation methodology based on fifteen statistical indicators and linking the SM RU and other statistical performance indicators, and 3) diagnosing the impact degrees of soil and vegetation environmental factors on SM products. Without any doubt, SM estimations are critical in arid and semiarid regions and are heavily affected by irrigation [7], [34]. With respect to the water budget, irrigation increases SM and in turn ET while decreasing runoff and depleting river and irrigation channel flows as well as groundwater, depending on the agricultural water use strategies [35]. From a hydrological perspective, irrigation strategies have huge impacts on local hydrological dynamics such as groundwater flow systems and groundwater-surface water storage changes, which would further affect streamflow, ET, and SM variations [36], [37]. Therefore, it is important to consider the complex water source use in SM validation in model simulations. Numerous validation studies based on simulated SM have been widely carried out, e.g., in the Rur and Erft catchments in Germany [38], over the US continent [39], on the Tibetan Plateau in China [40], and globally [41], [42]. In this study, we provide a comprehensive validation and intercomparison of ten SM products, namely, JAXA, LPRM-C, LPRM-X, SMOS-IC, CCI-A, CCI-P, CCI-C, GLEAM-A, GLEAM-B, and GLDAS SM products, based on outputs from a high-resolution integrated hydrologic model (ParFlow-CLM) in a typical managed agricultural region, the Heihe River Basin (HRB) in Northwestern China [Fig. 1(a)].

The objectives of this study are: 1) to evaluate ten SM products using integrated hydrologic model simulations based on fifteen statistical performance indicators, 2) to diagnose the impacts of eight environmental factors (DEM, clay content, sand content, porosity, saturated hydraulic conductivity, organic matter content, leaf area index, and root abundance) on SM products, and 3) to find the precise correlations between the relative uncertainties and other statistical performance indicators. The fifteen statistical performance indicators employed in this study include the mean absolute difference (MAD), mean bias difference (MBD), RMSE, standard deviation of the residual (SD), uncertainty at 95% (U95), the t-statistic (TS), Pearson's correlation coefficient (R-Pearson), Spearman's Rho correlation coefficient (R-Spearman), slope of best-fit line (SBF), Nash-Sutcliffe's efficiency (NSE), Legates's coefficient of efficiency (LCE), Willmotts's index of agreement (WIA), the Kolmogorov-Smirnov integral (KSI), a combined performance index (CPI), and the TCH-based RU.

The rest of this article is organized as follows. Section II introduces the study area, the integrated hydrologic model, multisource SM products, and other auxiliary data. Section III provides the validation methodologies and explains the

3



Fig. 1. ParFlow-CLM modeling domain and satellite pixel locations in the middle reaches of the HRB. (a) Location. (b) Grids and landcovers map. (c) Grids and hydrometeorological stations over DEM map. (d) TDRP and CRNP locations of the Daman superstation.

calculation approaches of impact quantifications and factor contributions. Section IV presents the assessment results, and Section V provides a further discussion on SM RU. Finally, conclusions are summarized in Section VI.

# II. STUDY AREA AND DATASETS

# A. Study Area and In Situ Observations

The Heihe River, the second largest inland river situated in arid-semiarid region of Northwestern China, originates from the alpine region of the Qilian Mountains, flowing north and terminating in the Juyan Lakes in the Gobi deserts [43]. The study area is located in the middle reaches of the HRB (hereafter referred as the mHRB,  $38.5^{\circ}$ – $39.5^{\circ}$  N and  $100^{\circ}$ – $101^{\circ}$  E), which is controlled by an endorheic climate with annual precipitation ranging from 100 to 200 mm and potential evaporation within the range of 1200–1800 mm [22], [44]. Being famous for its grain production, the mHRB is covered by a large amount of irrigated farmland [Fig. 1(b)]. Several irrigation canals have been built to maintain local agricultural demands. The elevation ranges from 1303 to 4065 m, with Zhangye Basin sitting in the central

section and Longshou Mountain located on the northeastern edge [Fig. 1(c)].

The in situ SM measurements within the mHRB were established during the HiWATER project, which was initialized in 2012 and designed to be a comprehensive ecohydrological experiment [21]. In this study, the SM data collected using the time domain reflectometry probe (TDRP) and cosmic ray neutron probes (CRNP) [22] were used for validation [Fig. 1(d)]. The TDRP automatically collected SM data at a depth of 2 cm and 10-min intervals since September 2012. The CRNP was installed in October 2012 and measured the intensity of ambient low-energy tertiary neutrons to calculate surface SM with a radius of 360 m and at an original 10-min interval. All the in situ data were resampled to daily time steps after standard quality control processes [45].

# B. Multisource SM Products

Ten global surface SM products were considered in this study for intercomparison, including AMSR, SMOS, and ESA CCI, as well as the land data assimilation products GLEAM and GLDAS. Although these SM products were sourced from

0.25°

 $0.25^{\circ}$ 

Product	Subproducts	Period (yyyy-mm)	Temporal resolution	Spatial resolution	Represent depth	Source	Ref.
ParFlow-CLM simulations	-	2010-01-2013-12	Hourly/daily	$0.005^{\circ}$	4.5 cm	This study	-
	JAXA	2002-07-2011-10				JAXA	[9, 46, 47]
AMSR	LPRM-X LPRM-C	and 2012-07 ongoing	Daily	0.25°	2 cm	NASA	[48]
SMOS	SMOS-IC	2010-11 ongoing	Daily	25 km	5 cm	CATDS	[51, 55]
	CCI-A	1991-08 ongoing	-				
ESA CCI	CCI-P	1978-11 ongoing	Daily	0.25°	3 cm	ESA	[14, 52]
	CCI-C	1978-11 ongoing	-				

Daily

3 h

TABLE I SUMMARY OF THE RS SM PRODUCTS AND PARFLOW-CLM SIMULATED REFERENCE USED FOR INTERCOMPARISON

different technologies (i.e., retrieval, reanalysis and merge, and assimilation), here they were collectively referred to as "RS SM products" and distinguished as single RS-source SM, merged SM, and assimilated SM, respectively. Information on the SM products is shown in Table I.

GLEAM-A

GLEAM-B

GLDAS-Noah

GLEAM

GLDAS

1980-01 ongoing

2003-01 ongoing

2000-01 ongoing

1) AMSR (AMSR-E/AMSR2): The AMSR SM products contain two series of products: the pioneer AMSR-E and its successor, AMSR2. The AMSR-E on the EOS Aqua satellite was launched by NASA on May 4, 2002 [8], while the AMSR2 onboard the GCOM-W1 satellite was launched by JAXA on May 18, 2012 [9]. As the ascending and descending overpasses of AMSR2 are identical to those of AMSR-E, the term "AMSR" represents the combination of both products.

Here, we utilized the three most widely used SM products from the AMSR series: JAXA SM product (version 8 for AMSR-E and version 3 for AMSR2, hereafter referred to as JAXA products) [9], [46], [47] and two LPRM SM products (generated from the C-band and X-band, hereafter referred to as LPRM-C and LPRM-X products, respectively) [48]. The JAXA SM data are available at https://www.eorc.jaxa.jp/AMSR/index\_en. html, while LPRM SM data can be found on the NASA Earth Observation Data website (https://search.earthdata.nasa.gov/).

2) SMOS IC: The SMOS satellite was launched on November 2, 2009, and has since become an invaluable tool for monitoring key components of the global water cycle. SMOS provides global coverage at the equator every three days, with an ascending orbit at 06:00 local solar time and a descending orbit at 18:00 local solar time [49]. The L-band microwave emission of the biosphere (L-MEB) model, based on a zero-order radiative transfer equation, is the forward model used for SMOS SM retrieval [50]. The newly developed SMOS-IC product (version 1.d) is a recently improved product that corrects some errors in the initialization of vegetation optical depth (version v.105) [51] The data can be accessed from https://data.catds.fr/cecsm/Land\_products/.

3) ESA CCI: This study utilized three ESA CCI SM datasets: the active RS product (referred to as the CCI-A product), which is generated by fusing four scatterometers based on C-band observations, the passive RS product (referred to as the CCI-P product), generated from ten radiometers, and the combined product (referred to as the CCI-C product), which is a blended product based on CCI-A and CCI-P results [52]. These datasets were originally developed by the University of Vienna and can be accessed from https://www.esa-soilmoisture-cci.org/.

10 cm

10 cm

VU

NASA

[53]

[15]

4) GLEAM: Two GLEAM SM datasets (version 3.6), GLEAM-A and GLEAM-B, were used in this study, which are generated from different forcing data in the surface layer (0–0.01 m) [53]. GLEAM-A uses radiation and air temperature datasets from the reanalysis ERA-Interim, while GLEAM-B is driven by RS datasets [16]. Furthermore, precipitation, snow water equivalent, vegetation optical depth, and vegetation fractions are largely driven by RS data for both datasets. Additionally, both GLEAM-A and GLEAM-B assimilate the CCI-C SM product. The dataset utilized in this study was obtained from Vrije University Amsterdam (VU) and can be accessed from https://www.gleam.eu/.

5) GLDAS: The GLDAS was provided by a joint effort of NASA's Goddard Space Flight Center and NOAA's National Centers for Environmental Prediction (NCEP). Among the four available land surface models, the level-4 Noah model [54] was selected. The GLDAS2 Noah SM data with 0.25° spatial resolution and a temporal resolution of 3 h were obtained, and the top layer (0–0.01 m depth) of the output SM was used. Daily averaged values were employed in this study.

# C. Auxiliary Data

The DEM and landcover maps are shown in Fig. 1. Spatial variation analysis was conducted using eight environmental factors, namely: 1) DEM; 2) clay content (hereafter referred to as clay); 3) sand content (hereafter referred to as sand); 4) porosity (Por); 5) saturated hydraulic conductivity (Ks); 6) organic matter content (OMC); 7) leaf area index (LAI); and 8) root abundance (RA). The LAI data were sourced from the "30 m month compositing leaf area index (LAI) product of the Heihe River Basin" [56] and can be downloaded from the National Tibetan Plateau/Third Pole Environment Data Center. Soil texture data utilized in this study were obtained from [57]. The Por, Ks, OMC, and RA data were sourced from two datasets, namely, "The Global Dataset of Soil Hydraulic and Thermal Parameters for Earth System Modeling" [58] and "The Global Soil Dataset for Earth System Modeling" [59] (http://globalchange.bnu.edu.cn/research/data). These four data

Classification	Parameter	Value			
	Domain	0.005°×0.005°×102 m			
_	Horizontal resolution	0.005°			
		Layer 1: 0.045 m			
		Layer 2: 0.046 m			
Model geometry		Layer 3: 0.075 m			
Woder geometry	Vortical resolution	Layer 4: 0.123 m			
	vertical resolution	Layer 5: 0.204 m			
		Layer 6: 0.336 m Layer 7: 0.554 m			
		Layer 8: 100.617 m			
	Saturated hydraulic conductivity	From [58, 71, 72]			
	Porosity	0.4			
Subgurfage by droulie	Specific storage	$10^{-5}$			
subsultace liyutaulic —	van Genuchten Parameters	_			
coefficients	Pore-size radius ( $\alpha$ )	$3.5 \text{ m}^{-1}$			
	Pore-size distribution ( <i>n</i> )	2.0			
	Res. vol. water content	0.015			
	Manning's roughness	$5.5 \times 10^{-6} \text{ m}^{-1/3} \text{ h}$			
Surface coefficients	Slopes	From [66]			
	Landcovers	From [67]			

TABLE II SUMMARY OF MODEL PARAMETERS USED IN PARFLOW-CLM SIMULATIONS

were also resampled to a spatial resolution of 0.005° to fit the ParFlow-CLM grid.

#### III. METHODOLOGY

#### A. ParFlow-CLM Model

The reference SM in this validation study was simulated by the integrated hydrologic model ParFlow-CLM [60], [61], which couples the ParFlow model for simulating groundwater [62], [63] and the Common Land Model (CLM) [64], and thus feasible for simulating regional SM with high-resolution. ParFlow-CLM solves the three-dimensional variably saturated groundwater flow and surface flow in a globally implicit manner; the land-surface formulation includes radiation budget, plant water use, land-energy balance, and snow processes. More information on the ParFlow-CLM can be found in Maxwell and Condon [65], and here, we only provide the water budget calculation as follows [28]:

$$P_{rain} + P_{snow} + R_{irri} + R_{in} - R_{out} + Q_{in} - Q_{out} - ET_{veg}$$
$$- E_{dir} = \Delta S_{soil} + \Delta S_{surf} + \Delta S_{gw} + \Delta S_{snow}$$
(1)

where  $P_{rain}$ ,  $P_{snow}$ ,  $R_{irri}$ ,  $R_{in}$ ,  $R_{out}$ ,  $ET_{veg}$ , and  $E_{dir}$ represent rain, snow, anthropogenic irrigation, inward overland flows, outward overland flow, subsurface influx, subsurface outflux, transpiration, and evaporation from barelands, respectively.  $Q_{in}$  and  $Q_{out}$  denote groundwater flux to deeper reservoirs and downstream from the mHRB.  $\Delta S_{soil}$ ,  $\Delta S_{surf}$ ,  $\Delta S_{gw}$  and  $\Delta S_{snow}$  represent the soil reservoir, surface water ponding, groundwater reservoirs, and snow water equivalent, respectively. All the components of the water budget in (1) are in unit volume per time. In this study, the ParFlow-CLM model was horizontally discretized at a  $0.005^{\circ}$  (-500 m) spatial resolution with 8 vertical layers with a total thickness of 102 m. The top layer was 0.045 m, and a free overland surface boundary condition was employed for continuity of pressure and flux at the surface–subsurface interface. The DEM data used to calculate *x*- and *y*-direction slopes was derived from the "30 m Aster-gdem Data in Qilian Mountain Area" [66]. The land cover map was obtained from the "Landcover dataset at Qilian Mountain area (V1.0)" [67]. The original spatial resolutions of both datasets were 30 m and then were resampled to the same criteria as those of the ParFlow-CLM. The model parameters, including van Genuchten parameters, Manning coefficients, subsurface geology properties, and vegetation properties, are listed in Table II.

The forcing data consists of the hydrometeorological data and agricultural water use data. The required atmospheric forcings of ParFlow-CLM (i.e., temperature, precipitation, specific humidity, longitudinal and latitudinal wind speeds, and downward longwave and shortwave radiations) were obtained from the gridded outputs by Weather Research and Forecasting (WRF) model simulations [68]. Considering the extensive usage and the associated effects of irrigation on the regional water and energy budgets, the agricultural water use including both surface water and groundwater irrigation, was derived from the Irrigation Datasets with 30 s spatial resolution over the entire HRB [69], [70]. Subsequently, irrigation is applied to the surface as spray irrigation. The irrigation schedule was made on an hourly basis throughout the growing seasons according to the Zhangye Water Affairs Bureau (www.zhangye.gov.cn/swj/). All the data could be accessed from the National Tibetan Plateau Data Center/Third Pole Environment Data Center (http://data.tpdc. ac.cn/).

The ParFlow-CLM simulation began with a 3-year spin-up (2008–2010) followed by a 3-year simulation (2011–2013). The initial time step was hourly based but outputted at daily scale. Differences between ET and I were calculated as well to obtain the response of SM products to I-ET.

# *B.* Validation Methodologies (Statistical Performance Indicators)

As previously noted, assessments and quantitative analyses of SM products and their behaviors are crucial for understanding the terrestrial water cycle. Previous studies have tended to focus on a particular aspect, such as qualitative analysis of deviation against observations. For decades, RS SM validation has relied on simple conventional statistical indicators, such as MAD, RMSE, and coefficient of determination [26], [29]. Comprehensive assessments are absolutely necessary by providing a multiclass evaluation. Therefore, in this study, we proposed an overall statistical indicator validation methodology to validate the SM [32].

The overall statistical indicator validation methodologies were first introduced in Gueymard [32] in solar product fields. Here, we revised the forms of validation indicators to better understand the overall performance of ten SM products. The statistical candidate indices and index metrics commonly employed in performance assessment can be classified into four categories as follows:

- 1) Indices representing the deviation (as for other model data) or error (as for in situ data) of individual points, of which the value would be 0 for a perfect match or ideal model.
- 2) Indices representing holistic performance (as for timeseries-self-correlation) or mutual coherence (as for crosscorrelation), of which the value would be 1 for a perfect match.
- 3) Indices representing distribution conformity, of which the value would be 0 for a perfect match.
- 4) Indices exploring the relative uncertainties.

More information about statistical indices and formulas can be found in [31] and [32].

In what follows, the ith simulated SM data (reference) point is noted as  $s_i$ , and the ith RS estimated SM data (target) point is noted as  $t_i$ . The average values of simulated SM data and RS estimated SM data are noted  $s_m$  and  $t_m$ , respectively. N is the number of the data points. The *i*th estimated-modeled difference (or error) is  $t_i - s_i$ . Unlike the absolute units used in some validation studies (e.g., [17], [30], [31], and [73]), the stating expression of the  $s_m$  value was converted back to the percent form, which is consistent with [32].

1) Class 1—indicators of dispersion: MAD, which is also referred as the mean absolute deviation or mean absolute error

$$MAD = \left(\sum_{i=1}^{i=N} |t_i - s_i|\right) / N.$$
(2)

MBD, which also refers to the mean bias deviation or mean bias error

$$MBD = \left[\sum_{i=1}^{i=N} (t_i - s_i)\right] / N.$$
(3)

RMSE is also referred to as the root mean square difference

RMSE = 
$$\left[\sum_{i=1}^{i=N} (t_i - s_i)^2 / N\right]^{1/2}$$
. (4)

SD is calculated as follows:

$$SD = \left[\sum_{i=1}^{i=N} N(t_i - s_i)^2 - \left\langle\sum_{i=1}^{i=N} (t_i - s_i)\right\rangle^2\right]^{1/2} / N.$$
 (5)

U95 is calculated as follows:

$$U_{95} = 1.96 (\mathrm{SD}^2 + \mathrm{RMSE}^2)^{1/2}.$$
 (6)

TS is calculated as follows:

$$TS = \left[ (N-1) \,\mathrm{MBD}^2 / \left( \mathrm{RMSE}^2 - \mathrm{MBD}^2 \right) \right]^{1/2}.$$
(7)

MAD, MBD, and RMSE are fundamental in RS retrieval and validation, while U95 and TS are less common in the RS field than former ones. Similarly, they convey relatively similar information, with the cosmetic advantage that a smaller value indicates a better SM retrieval.

2) Class 2—indicators of overall performance: R-Pearson is calculated as follows:

$$R - Pearson = \frac{\sum_{i=1}^{i=N} (t_i - t_m) (s_i - s_m)}{\left[\sum_{i=1}^{i=N} (t_i - t_m)^2 \sum_{i=1}^{i=N} (s_i - s_m)^2\right]^{1/2}}.$$
(8)

R-Spearman is calculated as follows:

R - Spearman = 1 - 6 
$$\sum_{i=1}^{i=N} (t_i - s_i)^2 / (N(N^2 - 1))$$
. (9)

SBF is calculated as follows:

$$SBF = \left[\sum_{i=1}^{i=N} (t_i - t_m) (s_i - s_m)\right] / \left[\sum_{i=1}^{i=N} (s_i - s_m)^2\right].$$
(10)

NSE is calculated as follows:

NSE = 1 - 
$$\left[\sum_{i=1}^{i=N} (t_i - s_i)^2\right] / \left[\sum_{i=1}^{i=N} (s_i - s_m)^2\right]$$
. (11)

LCE is calculated as follows:

$$LCE = 1 - \left[\sum_{i=1}^{i=N} |t_i - s_i|\right] / \left[\sum_{i=1}^{i=N} |s_i - s_m|\right].$$
 (12)

WIA is calculated as follows:

$$WIA = 1$$

$$-\left[\sum_{i=1}^{i=N} (t_i - s_i)^2\right] / \left[\sum_{i=1}^{i=N} (|t_i - s_m| + |s_i - s_m|)^2\right].$$
(13)

R-Pearson, R-Spearman, and SBF vary between 1 for a perfect positive match and -1 for a negative match, whereas 0 is total disagreement. LCE and NSE vary between 1 for perfect agreement and  $-\infty$  for complete disagreement. The WIA varies only between 1 and 0.

*3) Class 3—indicators of distribution similitude:* Here, the goal is to compare one or more cumulative frequency distributions of modeled data to those of a reference dataset. The KSI and CPI were utilized.

KSI is calculated as follows:

$$\mathrm{KSI} = \frac{100}{A_c} \int_{X_{\mathrm{min}}}^{X_{\mathrm{max}}} D_n d_x \tag{14}$$

where  $D_n$  is the absolute difference between the two normalized distributions within irradiance interval n.  $X_{\text{max}}$  and  $X_{\text{min}}$  are the maximum and minimum values of the binned reduced irradiance, x, and  $A_c$  is a characteristic quantity of the distribution

$$A_c = D_c \left( X_{\max} - X_{\min} \right) \tag{15}$$

where the critical value  $D_c$  is a statistical characteristic of the reference distribution, defined as a function of its number of points, N

$$D_{c} = \Phi(N) / N^{1/2}$$
(16)

where  $\Phi(N)$  is a pure function of N, for which an accurate numerical approximation has been obtained. Here, we assume that  $\Phi(N)$  is a constant  $\Phi(N) = 1.63$ .

CPI is defined as follows:

$$CPI = (KSI + RMSE)/2$$
(17)

where all values are expressed in the same unit as that of SM. KSI and CPI both vary between 0 for perfect agreement and  $\infty$  for complete disagreement. The cons and pros of the evaluation indicators and detailed guidelines can be found in [32].

4) Class 4—indicators of relative uncertainties: The TCH method facilitates the solution of estimating SM uncertainties with no prior knowledge. Fundamentally, the TCH method allows more than three SM products to take part in the calculation simultaneously, with a tolerance to the error cross-correlation. At a certain RS grid, TCH splits each SM time series of the observations into two terms

$$T_i = P_0 + \varepsilon_i, \ i = 1, \cdots, N \tag{18}$$

where  $P_0$  is the true SM value at the grid scale and  $\varepsilon_i$  represents the corresponding error. The TCH-based RU is

$$RU_i = \frac{\sqrt{r_{ii}}}{T_i}, \ i = 12, \cdots, N \tag{19}$$

where  $r_{ii}$  is the unknown diagonal element of the  $\varepsilon_i$  covariance.

Detailed methodologies of both TCH theory and TCH-derived RU equations are well documented in [31].

#### C. Definition of the Variation Coefficient

The variation coefficient (CV) is defined as follows:

$$CV = \frac{\sqrt{\sum_{i=1}^{i=n} (v_i - \bar{v})^2 / n}}{\bar{v}}, \ i = 1, 2, \cdots, n$$
 (20)

where  $v_i$  and  $\bar{v}$  are the ith, averaged value of SM and other environmental factors of 0.005° variables within a 0.25° grid. nis the sample number of the subgrids. Specifically, SM, DEM, Clay, Sand, Por, Ks, OMC, LAI, and RA participated in the CV calculations.

# D. Contribution of Environmental Factors to SM Uncertainty

The generalized additive model (GAM) was utilized to quantify the contributions of each factor to the total deviance of the ten SM RUs. GAM has been widely used in practice with its availability of statistically well-founded smoothing parameter estimation methods that are numerically efficient and robust [74]. Provided that  $g_i(RU_i)$  can be represented via several basis expansions of environmental factors ( $\psi_i$ ), the form of GAM is written as follows [75]:

$$g_i (RU_i) = \alpha + \sum_{i=1}^{i=M} f_i (\beta_i \psi_i) + \varepsilon$$
(21)

where  $RU_i$  is the TCH-based RU and  $g_i$  is the link function. M (M = 7 + 1) is the number of environmental factors along with the sample size.  $\alpha$  and  $\beta_i$  are unknown coefficients, and  $f_i$ are unknown basis functions used for smoothing environmental factors  $\psi_i$ .  $\varepsilon$  is the total error term. The GAM model (version 1.8-40) can be accessed from https://rdocumentation.org/packages/ mgcv/versions/1.8-40. For more details on the GAM algorithm, please refer to [74] and [75], while two examples in calculating contributions to RS SM can be found in [31] and [76].

#### **IV. RESULTS**

#### A. Validation of the ParFlow-CLM SM Simulation

The ParFlow-CLM simulated SM was first validated against two groups of in situ observations from the Daman meteorological superstation. The TDRP only probe a very small soil volume while the CRNP is better suited for the validation of satellite-based SM products due to larger footprint area (with a horizontal radius of about 360 m in the mHRB [45]). Fig. 2 displays the comparison of the ParFlow-CLM simulation with TDRP and CRNP SM observations for the Daman superstation. The observed SM data are the reference  $(s_i)$ , and the simulated SM data are the target  $(t_i)$ . The low RMSEs (< 0.06 m<sup>3</sup>/m<sup>3</sup>) and relatively high NSE indicated that the ParFlow-CLM simulation and observations are consistent in SM variations. The R-Person values were 0.83 and 0.93, while the R-Spearman values were 0.73 and 0.87 against TDRP and CRNP, respectively, demonstrating that the three SM time series were in good agreement. It is noted that ParFlow-CLM simulated results were almost equally distributed around the 1:1 line with TDRP and CRNP SM observations [Fig. 2(b)], demonstrating that the accuracy of the simulated SM. All the three SM time series can capture rainfall events with different magnitudes. Moreover, effects of irrigation on SM were identified through SM time series fluctuations from June to September. In winters, ParFlow-CLM SM continuously decline similar to observations. This showed that ParFlow-CLM simulation captured the temporal variation of both TDRP and CRNP SM measurements. All these revealed that ParFlow-CLM SM simulations agreed well with TDRP and CRNP observations and could fully reflect seasonality features of SM.



Fig. 2. Comparison of the ParFlow-CLM SM simulation (ParFlow-CLM Sim) with TDRP and CRNP SM observations (Daman TDRP Obs and Daman CRNP Obs) for the Daman superstation. (a) Three SM time series in 2013. (b) Scatter plot in the corresponding period. The unit of RMSE is  $m^3/m^3$ .

TABLE III Error Metrics of ten SM Products

	<b>N</b> 2 -	MAD	MBD	RMSE	SD	U95	TS	<b>R-Pearson</b>	R-Spearman	SBF	NSE	LCE	WIA	KSI	CPI	RU
Products	N0.2			Class 1					С	lass 2				Cla	ss 3	Class 4
JAXA	722	0.133	-0.133	0.145	0.057	0.305	61.8	0.438	0.692	0.17	- 0.90	- 0.23	0.51	0.95	0.54	35.12
LPRM-C	758	0.080	-0.033	0.095	0.089	0.255	10.2	- 0.191	- 0.292	-0.15	-0.38	-0.17	0.30	0.16	0.13	37.18
LPRM-X	758	0.082	-0.028	0.097	0.093	0.263	8.3	- 0.284	- 0.405	-0.26	-0.58	-0.24	0.26	0.13	0.11	34.73
SMOS-IC	317	0.118	-0.097	0.134	0.092	0.319	18.7	0.189	0.224	0.26	- 3.22	- 0.83	0.30	0.57	0.35	99.25
CCI-A	455	0.138	-0.133	0.148	0.064	0.316	43.6	0.460	0.542	0.35	- 0.65	- 0.38	0.39	0.64	0.39	53.36
CCI-P	538	0.112	-0.107	0.125	0.065	0.276	38.4	0.319	0.296	0.21	-0.38	-0.20	0.42	0.44	0.28	37.09
CCI-C	553	0.080	-0.079	0.091	0.044	0.198	41.1	0.445	0.442	0.14	0.04	0.07	0.34	0.35	0.22	13.18
GLEAM-A	1097	0.058	-0.026	0.061	0.066	0.176	15.6	0.230	0.147	0.16	0.22	0.18	0.58	0.25	0.15	28.73
GLEAM-B	1097	0.053	-0.031	0.055	0.057	0.155	22.6	0.443	0.294	0.30	0.36	0.26	0.67	0.32	0.18	22.93
GLDAS	1097	0.050	-0.041	0.053	0.050	0.142	40.4	0.619	0.601	0.27	0.26	0.22	0.54	0.41	0.23	9.19
Mean absolute d	lifference	(MAD),	mean bias differen	ce (MBD), R	MSE (roo	t mean squ	are error), S	D (standard deviation	on of the residual),	U95 (un-o	ertainty a	t 95%), and	d TS (t-sta	tistic) are catego	orized as Clas	s 1. R-Pearson

(Pearson's correlation coefficient), R-Spearman (Spearman's Rho correlation coefficient), SBF (slope of best-fit line), NSE (Nash-Sutcliffe's efficiency), LCE (Legates's coefficient of efficiency) and WIA (Willmotts's index of agreement) are categorized as Class 2. KSI (Kolmogorov-Smirnov Integral) and CPI (Combined Performance Index) are categorized as Class 3. The TCH-based RU is Class 3. Bold data in the table represent the best results concerning each error metric. Note that single-source RS SM products contain JAXA, LPRM-X, LPRM-C and SMOS SM products; merged SM are CCI-A, CCI-P, and CCI-C SM products; assimilated SM are GLEAM-A, GLEAM-B, and GLDAS SM products.

No. means the comparison data pairs of SM products in the validation period.

Compared with the TDRP SM time series data, the ParFlow-CLM SM had a tendency similar to that of CRNP, owing to different spatial scales. According to Montzka et al. [77], the CRNP approach is better suited for evaluations of satellite-based SM products because it integrates small-scale spatial variations in SM [78]. The validated results demonstrated that the SM simulated by ParFlow-CLM exhibited a higher degree of similarity to the SM dynamics obtained from the CRNP, along with better correlation. This suggests that the ParFlow-CLM SM simulation is capable of representing SM fluctuations at kilometer scale, thus providing an enhanced reference dataset for validating single-source RS, merged, and assimilated SM products. Notably, the time-series of the TDRP SM presented a discontinuity period, which could be attributed to the inclusion of extra hydrogen sources. This observation reinforces the potential of using hydrologic model simulations as a reliable reference dataset, exhibiting improved time-series continuity, for validating SM products [79], [80].

# B. Performance of Ten SM Products

All available SM estimates for each product during the whole period (2011–2013) were evaluated to obtain the accurate

statistical indicators for the purposes that users can utilize these SM products directly [17]. First, AMSR and SMOS RS SM data were resampled to daily time steps after quality control processes. GLDAS 3-h SM data were also merged to daily scale. All the results of the statistical performance indicators are listed in Table III.

1) Time-Series Comparisons and Performance Statistics: The time series for all SM products are plotted in Fig. 3(a). It is observed that the JAXA SM product exhibited a significantly limited dynamic range. Both LPRM-X and LPRM-C SM exhibited larger dynamic ranges compared with ParFlow-CLM simulations. Bindlish et al. [81] found that LPRM SM products have huge anomalous SM retrievals that are greater than the soil porosity over the Little Washita watershed, which was consistent with our study in the mHRB. Lu et al. [73] attributed this phenomenon to the erroneous estimation of the vegetation optical thickness. SMOS-IC SM had the largest variations with several discontinuities after quality controls. The three CCI SM products tended to underestimate the ParFlow-CLM SM to different extents. However, the distribution of CCI-C SM was close to that of ParFlow-CLM SM. The SM time-series data pairs of GLEAM-A, GLEAM-B, and GLADS were the same as ParFlow-CLM [see Fig. 3(d)] because they could all be regarded



Fig. 3. Comparisons of ParFlow-CLM SM and ten SM products. (a) Time series of the domain-averaged ParFlow-CLM SM and ten SM products during 2011–2013. (b) Scatter plot of ten SM products showing the overall performance. (c) Rose diagram of correlations of ten SM products with I-ET. (d) Bar graph showing data pairs of SM products involved in the comparisons.



Fig. 4. Spatial distribution patterns of eight environmental factors. (a) DEM. (b) Sand (sand content in %). (c) Clay (clay content in %). (d) Por (Porosity). (e) Ks (saturated hydraulic conductivity in cm/day). (f) MOC (Organic matter content). (g) LAI (Leaf area index). (h) RA (Root abundance).

as "modeled SM" [17]. Additionally, the averages and variances of these three SM products were similar to those of the ParFlow-CLM SM, with smaller ranges. The overall performance shown in Fig. 3(b) was based on R-Pearson calculated using all nine RS grid data. The LPRM-X, LPRM-C, GLEAM-A, GLEAM-B,

and GLDAS SM retrievals are near-equally distributed around the 1:1 line against the ParFlow-CLM data.

Table III shows the overall statistical indicators for the ten SM products against the ParFlow-CLM SM. For the deviation or error of SM time-series statistics (Class 1), GLDAS SM



Fig. 5. Impacts of spatial variations in SM estimation using heatmaps. (a) Biases of ten SM products against ParFlow-CLM on the nine  $0.25^{\circ}$  grids as well as the entire domain. (b) CVs (variation coefficients) of SM and environmental factors on the nine  $0.25^{\circ}$  grids as well as the entire domain. Note that the "SM" here is the original  $0.005^{\circ}$  ParFlow-CLM SM. "Max|10bias]" denotes the maximum value of the absolute bias of ten SM products. This value is plotted to better detect the connections between the environmental variable heterogeneities and SM biases.

had a low MAD value of 0.05 m<sup>3</sup>/m<sup>3</sup>, a small RMSE value of 0.054 m<sup>3</sup>/m<sup>3</sup>. The minimum U95 value was 0.142 m<sup>3</sup>/m<sup>3</sup> indicating that its SM points were highly close to ParFlow-CLM SM. Almost all the RS SM underestimated the ParFlow-CLM SM, with negative MBD values, while GLEAM-A SM had the smallest value ( $-0.026 \text{ m}^3/\text{m}^3$ ). The SD indicators showed that CCI-C performs the best since it had the lowest standard deviation with ParFlow-CLM SM (0.044 m<sup>3</sup>/m<sup>3</sup>). For an integrated statistic TS, LPRM-X depicted the smallest value of 8.3 m<sup>3</sup>/m<sup>3</sup>. In the second category statistics (Class 2), the NSE, LCE, and WIA values of GLEAM-B SM were the largest, which were 0.26, 0.22, and 0.54, respectively. Unlike these three efficiency coefficients, the CCI-A SM had the highest SBF with a value of 0.27. In the correlation statistics, GLDAS and JAXA SM had the largest R-Pearson and R-Spearman values of 0.619 and 0.601, respectively. In the third category statistics (Class 3), LPRM-X SM was the best performing SM product, with KSI and CPI values down to 0.16 and 0.13, respectively. In a statistical sense, the distributions of LPRM SM and ParFlow-CLM SM can be considered identical. Here we can find that the error metrics in Class 1 mainly show the deviations while statistics of Class 2 primarily depict the mutual coherence. Statistical indicators in Class 3 represent the entire distribution conformity.

For the TCH-based RU results (Class 4), GLDAS had the smallest value, while SMOS-IC obtained the largest value. Considering that GLEAM-B and GLDAS participated in RU calculations, the CCI-C SM gave the lowest RU value (13.18%) except for these two SM products involved. In summary, merged SM and assimilated SM products outperformed single RS-source SM products. These results were similar to SM validation using in situ networks (e.g., [82], [83], [84], [85], [86], and [87]). All the validation results concluded that JAXA SM shows underestimations, and LRPM SM shows overestimations in the mHRB. Additionally, the merged and assimilated SM outperformed the single-source RS SM products. In general, the GLEAM-B and GLDAS SM products exhibited higher levels of accuracy. Furthermore, assimilated or merged products demonstrated superior performance compared with individual RS-based SM products. The GLDAS SM product demonstrated the highest level of accuracy, which can be attributed to its extensive calibration using ground-based observations (https://disc.gsfc.nasa.gov/datasets/ GLDAS\_NOAH025\_3H\_2.1/). Additionally, the ESA and VU continuously release new versions of CCI and GLEAM SM products to achieve better precision. Gruber et al. [52] stated that one of the goals of the ESA CCI SM merging algorithm developments was to improve the performance statistics. Therefore, the single RS-source SM products need to be improved using more in situ observations and high temporal-spatial resolution simulated results (such as ParFlow-CLM) for evaluation and calibration.

In summary, for Class 1 (indices representing the deviation), CCI-C SM had the smallest SD value while GLEAM-A had the smallest MBD value. The TS value of LPRM-X was the smallest. The MAD, RMSE and U95 values of GLDAS SM were quite minimum. In indicators of Class 2 (representing holistic performance), the NSE, LCE, and WIA values of GLEAM-B SM were the largest. The R-Person and R-Spearman values of GLDAS SM were the largest while the CCI-A SM had the largest SBF value. In indicators of Class 3 (representing distribution conformity), LPRM-X SM had the smallest KSI and CPI values. Lastly for the Class 4 (TCH-based RU results), GLDAS SM had the smallest value.

2) Response of SM Products to I-ET: In addition to the temporally compared results presented in Fig. 3(a), correlations between SM and I-ET (infiltration - evapotranspiration) were investigated with daily step simulations. As shown in Fig. 3(a), ET varies significantly in the mHRB region, and the largest ET values occur during the monsoon season (from June to September). It was usually generated after precipitation events. We found that almost all the RS time-series SM products respond to variations in I-ET to different extents. Only LPRM-X and LPRM-C SM showed poor agreements with I-ET.

Fig. 3(c) shows the correlations between ten SM products and ParFlow-CLM I-ET simulation. JAXA had the highest R-Pearson/R-Spearman (0.423/0.496) among all the validated SM products, showing that it is the most sensitive to vertical moisture changes. Although the statistical performance indicators of JAXA showed poor agreement with ParFlow-CLM SM



Fig. 6. Deviance explained by LAI related to the TCH-based SM RU (relative uncertainties). (a) JAXA, SMOS and CCI-A. (b) CCI-P, CCI-C, GLEAM-A, GLEAM-B, and GLDAS. Dark lines are GAM effect splines. Grey shadow portion represents the middle 90% (5%–95%) of all samples.

variations, they were relatively more linked to temporal I-ET variability. Therefore, JAXA SM has strong application potential in moisture memory calculations and drought identifications [34], [88], such as a pulse-reserve indicator in the soil–plant continuum water balance [89]. Both LPRM-C and LPRM-X SM showed negative correlations with I-ET, meaning that LRPM SM products need to be improved in capturing the dynamics of vertical moisture movements. For the rest, all the merged or assimilated SM products (namely, CCI-A, CCI-P, CCI-C, GLEAM-A, GLEAM-B, and GLDAS) had a weak positive correlation with I-ET, which met basic expectations [6]. In summary, JAXA had high sensitivity to I-ET while LPRM-X and LPRM-C failed to capture the dynamics of I-ET.

#### C. Impacts of Environmental Factors on SM Products

Ks and porosity play an important role in the propagation of the head potential fluctuation in subsurface flow simulations which strongly affect the SM [90]. The soil texture and organic matter content factors are also sensitive to SM changes [81]. Meanwhile, the LAI and RA are significant vegetation factors in SM retrievals and assimilations [83], [91]. To understand the impact of each parameter on SM products, the spatial variations in the eight environmental factors (i.e., DEM, sand, clay, Por, Ks, OMC, LAI, and RA) were calculated, quantitatively estimated, and qualitatively analyzed. Fig. 4 maps the spatial distribution patterns of eight environmental factors. The null value was neglected in calculating the variation coefficients.

Fig. 5 presents the CVs of SM and environmental factors (b), together with the biases of SM products against ParFlow-CLM SM (a). Here, the term "Max|10bias]" denotes the maximum value of the absolute bias of ten SM products. This value is plotted to better detect the connections between the environmental variable heterogeneities and SM biases. Overall, the largest bias of the ten SM products against the ParFlow-CLM SM occurs in grids 2 and 5 [Fig. 5(a)]. Additionally, the greatest values of CVs in SM are generated in grids 2 and 5, indicating that the spatial variation is a nonnegligible element in RS SM.

Specifically, the CVs of Sand and Clay are relatively low (<0.9) and have small influences on SM variations. The CV of the DEM shows a one-way correlation with the CV of SM. That is, grids with large DEM CV values also have high SM CV values, indicating that DEM CV is an unnecessary but sufficient condition for SM CV. Por, OMC, and RA also have similar impacts as DEM on SM CV. Most importantly, the CVs of Ks and LAI have a strong correlation and consistency with SM CV. This indicates that Ks and LAI are dominant factors in SM spatial variations, which can represent the largest contributions of soil physical properties and overlying vegetation properties. Especially for Ks, the Ks CV values not only have a strong correlation with SM CV but also show consistency with the Max|10bias| values.

Overall, according to the impact patterns, we can classify the environmental factors into three categories (ranking from lowest to highest): 1) unnecessary and insufficient conditions, such as sand and clay; 2) unnecessary but sufficient conditions, such as DEM, Por, OMC, and RA; and 3) necessary and sufficient conditions, such as Ks and LAI. Previous studies (e.g., [82] and [84]) declared that the impacts of vegetation-related indices are generally larger than those of soil texture and that terrain complexity has an intervening impact. We expanded the involved environmental factors and provided their connections with SM products. Here we found that the variation coefficients of Ks and LAI had a strong correlation and consistency with the SM variation coefficient. This indicated that Ks and LAI, representing the contributions of soil physical properties and overlying vegetation properties, were dominant factors in SM spatial variations.

#### V. DISCUSSION

#### A. Relationship Between SM RU and Environmental Factors

In this study, we provide a comprehensive validation of ten multisource SM products against ParFlow-CLM simulations. A set of fifteen direct statistical performance indicators along with the TCH-based RU were utilized for the evaluations. These



Fig. 7. Scatter plots of TCH-based relative uncertainties (RU) and fourteen other statistical indicators. Lines are fitting curves using linear (for MAD, -MBD, RMSE, SD, U95, TS, SBF, NSE, LCE, WIA) and nonlinear (for R-Pearson, R-Spearman, KSI, CPI) fitting functions. Error bars represent the standard deviations of RU against other statistical indicators and plotted in vertical extension of each point. Fitting degree parameters with values of P-Person  $\geq 0.8$  and  $R^2 \geq 0.6$  are considered strong correlation relationships, and therefore corresponding parameters are marked in bold.

indicators have been commonly used in RS SM assessments during the past decades (e.g., [17], [20], [92], [93], [94], and [95]). However, the relative uncertainties were newly introduced to RS SM assessments [33], [96], [97], [98]. Subsequently, the key drivers for spatial variations of RS SM estimations were explored. Moreover, the linkages between SM RUs and other statistical performance indicators were discussed.

The drivers of eight environmental factors for TCH-based RU were explored using the GAM models. First, for a better balance in quantifying environmental factors, the relative elevation was picked by subtracting the minimum value of elevation. Then, the SM products were spatially resampled to obtain an SM RU matrix in ParFlow-CLM. Several screening procedures were conducted because the effects of sample size and data continuity must be carefully treated [31], [75]. The statistics of the first and second maximum contributing environmental factors are shown in Table IV. The GAM model was used to carry out the deviance explanation, which is related to ten SM relative uncertainties. The adjusted R<sup>2</sup> in each GAM model is also shown in Table IV. Both LPRM SM products are not included due to their adjusted

Products	First dominant	Contribution	Second dominant	Contribution	GAM R <sup>2</sup>	
	factor	fraction	factor	fraction		
JAXA	LAI	17.3%	DEM	14.9%	0.53	
SMOS-IC	Sand	11.4%	DEM	9.3%	0.45	
CCI-A	DEM	23.3%	OMC	17.9%	0.62	
CCI-P	LAI	17.8%	OMC	16.7%	0.63	
CCI-C	LAI	14.1%	DEM	11.1%	0.42	
GLEAM-A	LAI	15.8%	RA	5.3%	0.62	
GLEAM-B	LAI	12.3%	RA	5.6%	0.61	
GLDAS	LAI	10.1%	OMC	9.7%	0.37	

TABLE IV Statistics of the First and Second Maximum Contributing Environmental Factors (to TCH-Based SM Relative Uncertainties) and Their Contribution Fractions

 $R^2$  values being lower than 0.3. Results showed that LAI and RA had a decisive influence on SM RU. In particular, the LAI was the first dominant factor in almost all the SM products.

We find that the upper vegetation conditions, which include LAI and RA, have a decisive influence on SM RU accuracy. Almost all the merged and assimilated SM products are mainly impacted by LAI, possibly due to the anthropogenic influences, which include grain production, plantations, and wetland park development. All these human activities can strongly change LAI and therefore perturb SM variables. Therefore, in-depth analysis is needed to further explore the quantitative impacts of LAI on the TCH-based SM RU.

Fig. 6 displays the impacts of LAI on the TCH-based SM RUs, (a) for single RS-source SM products and (b) for merged SM and assimilated SM products. Specifically, the RU of JAXA initially decreased slightly and then exhibited an undulating increase until LAI reached 0.6. The RU of SMOS showed a decreasing trend before LAI reached 0.2, followed by a continuous increase thereafter. The RU of CCI-A exhibited a slight decrease and then remained constant, while the RU of CCI-P displayed a sharp decrease before starting to increase from LAI = 0.5. For CCI-C, the change pattern was similar to CCI-P with smaller magnitudes. In contrast, for assimilated SM products [GLEAM-A in blue, GLEAM-B green, GLDAS in pink in Fig. 6(b)], the RUs exhibited a monotonic decrease as LAI increased. This indicated that the impacts of LAI on these SM products were weaker than those of single RS-source SM products. This conclusion was similar to that of Liu et al. [31], which was obtained over the Qinghai-Tibet Plateau. Therefore, we can conclude that the impacts of upper vegetation layers on assimilated SM products are small. However, we can see that the RUs of GLEAM-A and GLEAM-B are larger than 10% when the LAI was smaller than 0.1 (under sparse vegetation or bare land).

# B. Linkage Between SM RU and Other Statistical Performance Indicators

Liu et al. [31] reported that the RUs and RMSE indices had good consistencies, which showed that the RUs share some mathematical connections with statistical performance indicators. Here, we performed a comprehensive analysis to find the precise relationships between the relative uncertainties and other statistical performance indicators. Fig. 7 maps the comparisons of indicators of Class 1, Class 2, Class 3, and RU. Scatter plots of TCH-based RU and fourteen other statistical indicators of ten SM products, which were obtained in performance validation, are used here to draw the panels. Here, we define P-Person  $\geq 0.8$  or  $R^2 \geq 0.6$  as an intrinsic strong correlation relationship. As shown in Fig. 7, MAD, U95, NSE, and CPI showed different types consistent with TCHbased RU. The MAD and U95 displayed a strong positive linear correlation with RU, while the NSE showed a strong negative linear correlation. Furthermore, the CPI exhibited a nonlinear correlation relationship with RU with polynomial fitting. This indicates that in a statistical sense, MAD, U95, NSE, and CPI can be utilized as substitutes for quantifying the SM RU.

#### VI. CONCLUSION

This study aimed to perform a comprehensive validation of ten SM products (namely, JAXA, LPRM-C, LPRM-X, SMOS-IC, CCI-A, CCI-P, CCI-C, GLEAM-A, GLEAM-B, and GLDAS) using a set of statistical performance indicators in the middle reaches of the HRB. Forced by hydrometeorological and agricultural water use reanalysis data (including both surface water and groundwater irrigations), an integrated hydrologic model (ParFlow-CLM) was built to provide the reference SM data because of its superior performance in simulating SM [28], [99], [100]. Ten SM products were evaluated against the ParFlow-CLM simulations by a comprehensive evaluation framework composed of fifteen statistical performance indicators, which was classified into four categories as indicators of dispersion, overall performance, distribution similitude, and RU. Moreover, eight environmental factors (DEM, clay content, sand content, porosity, saturated hydraulic conductivity, organic matter content, leaf area index, and root abundance) were utilized to explore the drivers for spatial variations of these SM products. The precise correlations between the relative uncertainties and other statistical performance indicators were also investigated. The main findings are summarized as follows:

 The ParFlow-CLM simulation and SM observations were observed to be in reasonable agreement for a holistic performance. The R-Person values were 0.83 and 0.93, while the R-Spearman values were 0.73 and 0.87 against TDRP and CRNPs, respectively, indicating that the simulation and observations match well with the magnitude of the SM variations. Moreover, compared with TDRP SM time series, ParFlow-CLM SM was closer to that of CRNP, owing to different spatial scales. This indicated that ParFlow-CLM SM simulation can capture SM variations at spatial scale and is a better reference data for validating SM products.

- 2) For Class 1 (indices representing the deviation), CCI-C SM had the smallest SD value while GLEAM-A had the smallest MBD value. The TS value of LPRM-X was the smallest. The MAD, RMSE and U95 values of GLDAS SM were quite minimum. In indicators of Class 2 (representing holistic performance), the NSE, LCE, and WIA values of GLEAM-B SM were the largest. The R-Person and R-Spearman values of GLDAS SM were the largest while the CCI-A SM had the largest SBF value. In indicators of Class 3 (representing distribution conformity), LPRM-X SM had the smallest KSI and CPI values. Lastly for the Class 4 (TCH-based RU results), GLDAS SM had the smallest value.
- 3) In general, merged SM and assimilated SM outperform single-source RS SM products. JAXA SM had a very low dynamic range, while SMOS-IC SM had the largest variations. The accuracy of LPRM-X SM was higher than that of LPRM-C SM. The CCI-A, CCI-C, GLEAM-A, GLEAM-B, and GLDAS SM products reflect the seasonal variations in the mHRB very well with comparably high temporal correlations over the dense vegetated regions. Overall, the single RS-source SM products need to be improved using more in situ observations and high temporalspatial resolution simulated results for evaluation and calibration.
- 4) JAXA has the highest R-Pearson/R-Spearman (0.423/ 0.496) among all the validated SM products, showing that it is the most sensitive to vertical moisture changes. Although the statistical performance indicators of JAXA show poor agreement with ParFlow-CLM SM variations, they are relatively more linked to temporal infiltration-ET variability. Therefore, JAXA SM has strong application potential in moisture memory calculations and drought identifications.
- 5) The variation coefficients of saturated hydraulic conductivity (Ks) and leaf area index (LAI) have a strong correlation and consistency with SM variation coefficient. This indicates that Ks and LAI, representing the contributions of soil physical properties and overlying vegetation properties, are dominant factors in SM spatial variations. In particular, based on the GAM, the LAI is the most dominant factor with the largest contribution fraction to SM RU, possibly due to the anthropogenic influences. Moreover, deviances explained by LAI related to SM relative uncertainties in merged SM and assimilated SM are relatively small, demonstrating that the single-source RS SM products need to be improved over dense vegetated regions.
- Comparisons of indicators of Class 1 (representing the deviation), Class 2 (representing holistic performance), Class 3 (representing distribution conformity), and SM

RU show that the MAD and U95 have strong positive linear correlations with RU, while the NSE shows a strong negative linear correlation. Additionally, the CPI depicts a nonlinear correlation relationship with RU with polynomial fitting. In total and in a statistical sense, MAD, U95, NSE, and CPI can be utilized as substitutes for quantifying the SM RU.

The study domain (mHRB) was a small-sized, semiarid subbasin, with an alluvial aquifer that serves as a representative example of other arid and semiarid basins. Although the current study might be transferable to other regions, future work is essential to simulate larger climatic and hydrologic areas to better understand the performances of SM products. Future simulations of enlarged domains would also facilitate further analysis of the impacts of soil and vegetation factors on SM products as well as the linkage between SM RU and other statistical performance indicators.

#### ACKNOWLEDGMENT

The authors would like to thank the high-performance computing support from the Center for Geodata and Analysis, Faculty of Geographical Science, Beijing Normal University (https://gda.bnu.edu.cn/), and also the National Tibetan Plateau Data Center for data support.

#### REFERENCES

- [1] J. Wigneron, A. Chanzy, J. Calvet, and N. Bruguier, "A simple algorithm to retrieve soil moisture and vegetation biomass using passive microwave measurements over crop fields," *Remote Sens. Environ.*, vol. 51, no. 3, pp. 331–341, 1995, doi: 10.1016/0034-4257(94)00081-W.
- [2] K. L. Findell, P. Gentine, B. R. Lintner, and C. Kerr, "Probability of afternoon precipitation in eastern United States and Mexico enhanced by high evaporation," *Nature Geosci.*, vol. 4, no. 7, pp. 434–439, 2011, doi: 10.1038/ngeo1174.
- [3] P. Greve, B. Orlowsky, B. Mueller, J. Sheffield, M. Reichstein, and S. I. Seneviratne, "Global assessment of trends in wetting and drying over land," *Nature Geosci.*, vol. 7, no. 10, pp. 716–721, 2014, doi: 10.1038/ngeo2247.
- [4] X. Liu, X. Feng, and B. Fu, "Changes in global terrestrial ecosystem water use efficiency are closely related to soil moisture," *Sci. Total Environ.*, vol. 698, 2020, Art. no. 134165, doi: 10.1016/j.scitotenv.2019.134165.
- [5] T. Zhao et al., "Retrievals of soil moisture and vegetation optical depth using a multi-channel collaborative algorithm," *Remote Sens. Environ.*, vol. 257, 2021, Art. no. 112321, doi: 10.1016/j.rse.2021.112321.
- [6] C. Ma, X. Li, J. Wang, C. Wang, Q. Duan, and W. Wang, "A comprehensive evaluation of microwave emissivity and brightness temperature sensitivities to soil parameters using qualitative and quantitative sensitivity analyses," *IEEE Trans. Geosci. Remote Sens.*, vol. 55, no. 2, pp. 1025–1038, Feb. 2017, doi: 10.1109/TGRS.2016.2618903.
- [7] Z. Wang, T. Che, and Y. Liou, "Global sensitivity analysis of the L-MEB model for retrieving soil moisture," *IEEE Trans. Geosci. Remote Sens.*, vol. 54, no. 5, pp. 2949–2962, May 2016, doi: 10.1109/TGRS.2015.2509176.
- [8] E. G. Njoku, T. J. Jackson, V. Lakshmi, T. K. Chan, and S. V. Nghiem, "Soil moisture retrieval from AMSR-E," *IEEE Trans. Geosci. Remote Sens.*, vol. 41, no. 2, pp. 215–229, Feb. 2003, doi: 10.1109/TGRS.2002.808243.
- [9] K. Imaoka et al., "Global change observation mission (GCOM) for monitoring carbon, water cycles, and climate change," *Proc. IEEE*, vol. 98, no. 5, pp. 717–734, May 2010, doi: 10.1109/JPROC.2009.2036869.
- [10] Y. H. Kerr et al., "The SMOS mission: New tool for monitoring key elements of the global water cycle," *Proc. IEEE*, vol. 98, no. 5, pp. 666–687, May 2010, doi: 10.1109/JPROC.2010.2043032.

- [11] T. Zhao et al., "Soil moisture retrievals using L-band radiometry from variable angular ground-based and airborne observations," *Remote Sens. Environ.*, vol. 248, 2020, Art. no. 111958, doi: 10.1016/j.rse.2020.111958.
- [12] J. P. Wigneron et al., "Modelling the passive microwave signature from land surfaces: A review of recent results and application to the L-band SMOS & SMAP soil moisture retrieval algorithms," *Remote Sens. Environ.*, vol. 192, pp. 238–262, 2017, doi: 10.1016/ j.rse.2017.01.024.
- [13] W. Preimesberger, T. Scanlon, C. Su, A. Gruber, and W. Dorigo, "Homogenization of structural breaks in the global ESA CCI soil moisture multisatellite climate data record," *IEEE Trans. Geosci. Remote Sens.*, vol. 59, no. 4, pp. 2845–2862, Apr. 2021, doi: 10.1109/TGRS.2020.3012896.
- [14] W. Dorigo et al., "ESA CCI soil moisture for improved Earth system understanding: State-of-the art and future directions," *Remote Sens. Environ.*, vol. 203, pp. 185–215, 2017, doi: 10.1016/j.rse.2017.07.001.
- [15] M. Rodell et al., "The global land data assimilation system," Bull. Amer. Meteorological Soc., vol. 85, no. 3, pp. 381–394, 2004, doi: 10.1175/BAMS-85-3-381.
- [16] D. G. Miralles, T. R. H. Holmes, R. A. M. De Jeu, J. H. Gash, A. G. C. A. Meesters, and A. J. Dolman, "Global land-surface evaporation estimated from satellite-based observations," *Hydrol. Earth Syst. Sci.*, vol. 15, no. 2, pp. 453–469, 2011, doi: 10.5194/hess-15-453-2011.
- [17] J. Zheng et al., "Assessment of 24 soil moisture datasets using a new in situ network in the Shandian River Basin of China," *Remote Sens. Environ.*, vol. 271, 2022, Art. no. 112891, doi: 10.1016/j.rse.2022.112891.
- [18] A. Gruber et al., "Validation practices for satellite soil moisture retrievals: What are (the) errors?," *Remote Sens. Environ.*, vol. 244, 2020, Art. no. 111806, doi: 10.1016/j.rse.2020.111806.
- [19] T. Zhao et al., "Soil moisture experiment in the Luan River supporting new satellite mission opportunities," *Remote Sens. Environ.*, vol. 240, 2020, Art. no. 111680, doi: 10.1016/j.rse.2020.111680.
- [20] L. Brocca, F. Melone, T. Moramarco, and R. Morbidelli, "Spatialtemporal variability of soil moisture and its estimation across scales," *Water Resour. Res.*, vol. 46, no. 2, 2010, Art. no. W02516, doi: 10.1029/2009WR008016.
- [21] X. Li et al., "Heihe watershed allied telemetry experimental research (HiWATER): Scientific objectives and experimental design," *Bull. Amer. Meteorological Soc.*, vol. 94, no. 8, pp. 1145–1160, 2013, doi: 10.1175/BAMS-D-12-00154.1.
- [22] S. Liu et al., "The Heihe integrated observatory network: A basin-scale land surface processes observatory in China," *Vadose Zone J.*, vol. 17, no. 1, 2018, Art. no. 180072, doi: 10.2136/vzj2018.04.0072.
- [23] B. N. O. Kuffour, N. B. Engdahl, C. S. Woodward, L. E. Condon, S. Kollet, and R. M. Maxwell, "Simulating coupled surface-subsurface flows with parflow v3.5.0: Capabilities, applications, and ongoing development of an open-source, massively parallel, integrated hydrologic model," *Geoscientific Model Develop.*, vol. 13, no. 3, pp. 1373–1397, 2020, doi: 10.5194/gmd-13-1373-2020.
- [24] L. E. Condon, A. L. Atchley, and R. M. Maxwell, "Evapotranspiration depletes groundwater under warming over the contiguous United States," *Nature Commun.*, vol. 11, no. 1, 2020, Art. no. 873, doi: 10.1038/s41467-020-14688-0.
- [25] P. López López, E. H. Sutanudjaja, J. Schellekens, G. Sterk, and M. F. P. Bierkens, "Calibration of a large-scale hydrological model using satellite-based soil moisture and evapotranspiration products," *Hydrol. Earth Syst. Sci.*, vol. 21, no. 6, pp. 3125–3144, 2017, doi: 10.5194/hess-21-3125-2017.
- [26] T. J. Jackson et al., "Validation of soil moisture and ocean salinity (SMOS) soil moisture over watershed networks in the U.S.," *IEEE Trans. Geosci. Remote Sens.*, vol. 50, no. 5, pp. 1530–1543, May 2012, doi: 10.1109/TGRS.2011.2168533.
- [27] I. Jemcov and M. Petric, "Measured precipitation vs. effective infiltration and their influence on the assessment of karst systems based on results of the time series analysis," *J. Hydrol.*, vol. 379, no. 3-4, pp. 304–314, 2009, doi: 10.1016/j.jhydrol.2009.10.016.
- [28] M. M. F. O'Neill, D. T. Tijerina, L. E. Condon, and R. M. Maxwell, "Assessment of the ParFlow–CLM CONUS 1.0 integrated hydrologic model: Evaluation of hyper-resolution water balance components across the contiguous United States," *Geoscientific Model Develop.*, vol. 14, no. 12, pp. 7223–7254, 2021, doi: 10.5194/gmd-14-7223-2021.
- [29] T. J. Jackson et al., "Validation of advanced microwave scanning radiometer soil moisture products," *IEEE Trans. Geosci. Remote Sens.*, vol. 48, no. 12, pp. 4256–4272, Dec. 2010, doi: 10.1109/TGRS.2010.2051035.

- [30] J. Liu et al., "Evaluation of SMAP, SMOS-IC, FY3B, JAXA, and LPRM soil moisture products over the Qinghai–Tibet Plateau and its surrounding areas," *Remote Sens.*, vol. 11, no. 7, 2019, Art. no. 792, doi: 10.3390/rs11070792.
- [31] J. Liu et al., "Uncertainty analysis of eleven multisource soil moisture products in the third pole environment based on the three-corned hat method," *Remote Sens. Environ.*, vol. 255, 2021, Art. no. 112225, doi: 10.1016/j.rse.2020.112225.
- [32] C. A. Gueymard, "A review of validation methodologies and statistical performance indicators for modeled solar radiation data: Towards a better bankability of solar projects," *Renewable Sustain. Energy Rev.*, vol. 39, pp. 1024–1034, 2014, doi: 10.1016/j.rser.2014.07.117.
- [33] T. Xu et al., "Evaluation of twelve evapotranspiration products from machine learning, remote sensing and land surface models over conterminous United States," *J. Hydrol.*, vol. 578, 2019, Art. no. 124105, doi: 10.1016/j.jhydrol.2019.124105.
- [34] K. A. McColl, S. H. Alemohammad, R. Akbar, A. G. Konings, S. Yueh, and D. Entekhabi, "The global distribution and dynamics of surface soil moisture," *Nature Geosci.*, vol. 10, no. 2, pp. 100–104, 2017, doi: 10.1038/ngeo2868.
- [35] L. E. Condon and R. M. Maxwell, "Feedbacks between managed irrigation and water availability: Diagnosing temporal and spatial patterns using an integrated hydrologic model," *Water Resour. Res.*, vol. 50, no. 3, pp. 2600–2616, 2014, doi: 10.1002/2013WR014868.
- [36] I. E. M. de Graaf, T. Gleeson, L. P. H. Rens Van Beek, E. H. Sutanudjaja, and M. F. P. Bierkens, "Environmental flow limits to global groundwater pumping," *Nature*, vol. 574, no. 7776, pp. 90–94, 2019, doi: 10.1038/s41586-019-1594-4.
- [37] F. Felfelani, D. M. Lawrence, and Y. Pokhrel, "Representing intercell lateral groundwater flow and aquifer pumping in the community land model," *Water Resour. Res.*, vol. 57, no. 1, 2021, Art. no. e2020WR027531, doi: 10.1029/2020WR027531.
- [38] K. Rötzer et al., "Catchment scale validation of SMOS and AS-CAT soil moisture products using hydrological modeling and temporal stability analysis," *J. Hydrol.*, vol. 519, pp. 934–946, 2014, doi: 10.1016/j.jhydrol.2014.07.065.
- [39] T. W. Ford and S. M. Quiring, "Comparison of contemporary in situ, model, and satellite remote sensing soil moisture with a focus on drought monitoring," *Water Resour. Res.*, vol. 55, no. 2, pp. 1565–1582, 2019, doi: 10.1029/2018WR024039.
- [40] C. Li et al., "The evaluation of SMAP enhanced soil moisture products using high-resolution model simulations and in-situ observations on the Tibetan Plateau," *Remote Sens.*, vol. 10, no. 4, 2018, Art. no. 535, doi: 10.3390/rs10040535.
- [41] A. Al-Yaari et al., "Global-scale evaluation of two satellite-based passive microwave soil moisture datasets (SMOS and AMSR-E) with respect to land data assimilation system estimates," *Remote Sens. Environ.*, vol. 149, pp. 181–195, 2014, doi: 10.1016/j.rse.2014.04.006.
- [42] F. Chen et al., "Global-scale evaluation of SMAP, SMOS and ASCAT soil moisture products using triple collocation," *Remote Sens. Environ.*, vol. 214, pp. 1–13, 2018, doi: 10.1016/j.rse.2018.05.008.
- [43] X. Li et al., "Hydrological cycle in the Heihe River Basin and its implication for water resource management in endorheic basins," *J. Geophysical Res.: Atmos.*, vol. 123, no. 2, pp. 890–914, 2018, doi: 10.1002/2017JD027889.
- [44] G. Cheng et al., "Integrated study of the water–ecosystem–economy in the Heihe River Basin," *Nat. Sci. Rev.*, vol. 1, no. 3, pp. 413–428, 2014, doi: 10.1093/nsr/nwu017.
- [45] Z. Zhu, L. Tan, S. Gao, and Q. Jiao, "Observation on soil moisture of irrigation cropland by cosmic-ray probe," *IEEE Geosci. Remote Sens. Lett.*, vol. 12, no. 3, pp. 472–476, Mar. 2015, doi: 10.1109/LGRS.2014.2346784.
- [46] H. J. A. E. Fujii, T. Koike, and K. Imaoka, "Improvement of the AMSR-E algorithm for soil moisture estimation by introducing a fractional vegetation coverage dataset derived from MODIS data," *J. Remote Sens. Soc. Jpn.*, vol. 29, no. 1, pp. 282–292, 2009, doi: 10.11440/rssj.29.282.
- [47] H. T. U. J. Lu, T. Koike, H. Fujii, T. Ohta, and K. Tamagawa, "Development of a physically-based soil moisture retrieval algorithm for spaceborne passive microwave radiometers and its application to AMSR-E," *J. Remote Sens. Soc. Jpn.*, vol. 29, no. 1, pp. 253–262, 2009, doi: 10.11440/rssj.29.253.
- [48] M. Owe, R. de Jeu, and T. Holmes, "Multisensor historical climatology of satellite-derived global land surface moisture," *J. Geophysical Res.*, vol. 113, no. F1, 2008, Art. no. F01002, doi: 10.1029/2007JF000769.

- [49] Y. H. Kerr et al., "The SMOS soil moisture retrieval algorithm," *IEEE Trans. Geosci. Remote Sens.*, vol. 50, no. 5, pp. 1384–1403, May 2012, doi: 10.1109/TGRS.2012.2184548.
- [50] J. P. Wigneron et al., "L-band microwave emission of the biosphere (L-MEB) model: Description and calibration against experimental data sets over crop fields," *Remote Sens. Environ.*, vol. 107, no. 4, pp. 639–655, 2007, doi: 10.1016/j.rse.2006.10.014.
- [51] R. Fernandez-Moran et al., "A new calibration of the effective scattering albedo and soil roughness parameters in the SMOS SM retrieval algorithm," *Int. J. Appl. Earth Observ. Geoinf.*, vol. 62, pp. 27–38, 2017, doi: 10.1016/j.jag.2017.05.013.
- [52] A. Gruber, T. Scanlon, R. van der Schalie, W. Wagner, and W. Dorigo, "Evolution of the ESA CCI soil moisture climate data records and their underlying merging methodology," *Earth Syst. Sci. Data*, vol. 11, no. 2, pp. 717–739, 2019, doi: 10.5194/essd-11-717-2019.
- [53] B. Martens et al., "GLEAM v3: Satellite-based land evaporation and root-zone soil moisture," *Geoscientific Model Develop.*, vol. 10, no. 5, pp. 1903–1925, 2017, doi: 10.5194/gmd-10-1903-2017.
- [54] F. Chen et al., "Modeling of land surface evaporation by four schemes and comparison with FIFE observations," J. Geophysical Res.: Atmos., vol. 101, no. D3, pp. 7251–7268, 1996, doi: 10.1029/95JD 02165.
- [55] R. Fernandez-Moran et al., "SMOS-IC: An alternative SMOS soil moisture and vegetation optical depth product," *Remote Sens.*, vol. 9, no. 5, 2017, Art. no. 457, doi: 10.3390/rs9050457.
- [56] J. Zhao et al., "Leaf area index retrieval combining HJ1/CCD and Landsat8/OLI data in the Heihe River Basin, China," *Remote Sens.*, vol. 7, no. 6, pp. 6862–6885, 2015.
- [57] Z. Wang and W. Shi, "Robust variogram estimation combined with isometric log-ratio transformation for improved accuracy of soil particle-size fraction mapping," *Geoderma*, vol. 324, pp. 56–66, 2018, doi: 10.1016/j.geoderma.2018.03.007.
- [58] Y. Dai, W. Shangguan, Q. Duan, B. Liu, S. Fu, and G. Niu, "Development of a China dataset of soil hydraulic parameters using pedotransfer functions for land surface modeling," *J. Hydrometeorology*, vol. 14, no. 3, pp. 869–887, 2013, doi: 10.1175/JHM-D-12-0149.1.
- [59] W. Shangguan, Y. Dai, Q. Duan, B. Liu, and H. Yuan, "A global soil data set for Earth system modeling," J. Adv. Model. Earth Syst., vol. 6, no. 1, pp. 249–263, 2014.
- [60] R. M. Maxwell and N. L. Miller, "Development of a coupled land surface and groundwater model," *J. Hydrometeorology*, vol. 6, no. 3, pp. 233–247, 2005, doi: 10.1175/JHM422.1.
- [61] S. J. Kollet and R. M. Maxwell, "Capturing the influence of groundwater dynamics on land surface processes using an integrated, distributed watershed model," *Water Resour. Res.*, vol. 44, no. 2, 2008, Art. no. W02402, doi: 10.1029/2007WR006004.
- [62] S. J. Kollet and R. M. Maxwell, "Integrated surface-groundwater flow modeling: A free-surface overland flow boundary condition in a parallel groundwater flow model," *Adv. Water Resour.*, vol. 29, no. 7, pp. 945–958, 2006, doi: 10.1016/j.advwatres.2005.08.006.
- [63] R. M. Maxwell, "A terrain-following grid transform and preconditioner for parallel, large-scale, integrated hydrologic modeling," *Adv. Water Resour.*, vol. 53, pp. 109–117, 2013, doi: 10.1016/j.advwatres.2012. 10.001.
- [64] Y. Dai et al., "The common land model," Bull. Amer. Meteorological Soc., vol. 84, no. 8, pp. 1013–1024, 2003.
- [65] R. M. Maxwell and L. E. Condon, "Connections between groundwater flow and transpiration partitioning," *Science*, vol. 353, no. 6297, pp. 377–380, 2016, doi: 10.1126/science.aaf7891.
- [66] Y. Qi, J. Zhang, S. Zhou, and H. Wang, "30m aster-gdem data in Qilian Mountain Area (2018)," in *National Tibetan Plateau Data Center*, 2019, doi: 10.11888/Geogra.tpdc.270148.
- [67] B. Zhong et al., "Finer resolution land-cover mapping using multiple classifiers and multisource remotely sensed data in the Heihe River Basin," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 8, no. 10, pp. 4973–4992, Oct. 2015, doi: 10.1109/JSTARS.2015. 2461453.
- [68] X. Pan, X. Li, G. Cheng, H. Li, and X. He, "Development and evaluation of a River-Basin-scale high spatio-temporal precipitation data set using the WRF model: A case study of the Heihe River Basin," *Remote Sens.*, vol. 7, no. 7, pp. 9230–9252, 2015, doi: 10.3390/rs70709230.
- [69] Y. Zeng et al., "Effects of anthropogenic water regulation and groundwater lateral flow on land processes," J. Adv. Model. Earth Syst., vol. 8, no. 3, pp. 1106–1131, 2016, doi: 10.1002/2016MS000646.

- [70] Y. Zeng et al., "Ecohydrological effects of stream-aquifer water interaction: A case study of the Heihe River Basin, Northwestern China," *Hydrol. Earth Syst. Sci.*, vol. 20, no. 6, pp. 2333–2352, 2016, doi: 10.5194/hess-20-2333-2016.
- [71] W. Shangguan et al., "A China data set of soil properties for land surface modeling," J. Adv. Model. Earth Syst., vol. 5, no. 2, pp. 212–224, 2013, doi: 10.1002/jame.20026.
- [72] T. Gleeson, N. Moosdorf, J. Hartmann, and L. P. H. van Beek, "A glimpse beneath Earth's surface: GLobal HYdrogeology MaPS (GLHYMPS) of permeability and porosity," *Geophysical Res. Lett.*, vol. 41, no. 11, pp. 3891–3898, 2014, doi: 10.1002/2014GL059856.
- [73] Z. Lu et al., "Estimating time series soil moisture by applying recurrent nonlinear autoregressive neural networks to passive microwave data over the Heihe River Basin, China," *Remote Sens.*, vol. 9, no. 6, 2017, Art. no. 574, doi: 10.3390/rs9060574.
- [74] S. N. Wood, "Fast stable restricted maximum likelihood and marginal likelihood estimation of semiparametric generalized linear models," *J. Roy. Stat. Soc. Ser. B*, vol. 73, no. 1, pp. 3–36, 2011, doi: 10.1111/j.1467-9868.2010.00749.x.
- [75] S. N. Wood, N. Pya, and B. Säfken, "Smoothing parameter and model selection for general smooth models," *J. Amer. Stat. Assoc.*, vol. 111, no. 516, pp. 1548–1563, 2016, doi: 10.1080/01621459.2016.1180986.
- [76] Z. Hu et al., "Applying a wavelet transform technique to optimize general fitting models for SM analysis: A case study in downscaling over the Qinghai–Tibet Plateau," *Remote Sens.*, vol. 14, no. 13, 2022, Art. no. 3063, doi: 10.3390/rs14133063.
- [77] C. Montzka et al., "Validation of spaceborne and modelled surface soil moisture products with cosmic-ray neutron probes," *Remote Sens.*, vol. 9, no. 2, 2017, Art. no. 103, doi: 10.3390/rs9020103.
- [78] J. Dong, T. E. Ochsner, M. Zreda, M. H. Cosh, and C. B. Zou, "Calibration and validation of the COSMOS rover for surface soil moisture measurement," *Vadose Zone J.*, vol. 13, no. 4, pp. 1–8, 2014, doi: 10.2136/vzj2013.08.0148.
- [79] C. Montzka et al., "Brightness temperature and soil moisture validation at different scales during the SMOS validation campaign in the Rur and Erft catchments, Germany," *IEEE Trans. Geosci. Remote Sens.*, vol. 51, no. 3, pp. 1728–1743, Mar. 2013, doi: 10.1109/TGRS. 2012.2206031.
- [80] X. Fan, Y. Lu, Y. Liu, T. Li, S. Xun, and X. Zhao, "Validation of multiple soil moisture products over an intensive agricultural region: Overall accuracy and diverse responses to precipitation and irrigation events," *Remote Sens.*, vol. 14, no. 14, 2022, Art. no. 3339, doi: 10.3390/rs14143339.
- [81] R. Bindlish et al., "GCOM-W AMSR2 soil moisture product validation using core validation sites," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 11, no. 1, pp. 209–219, Jan. 2018, doi: 10.1109/JS-TARS.2017.2754293.
- [82] X. Song, J. Ma, X. Li, P. Leng, F. Zhou, and S. Li, "First results of estimating surface soil moisture in the vegetated areas using ASAR and hyperion data: The Chinese Heihe River Basin case study," *Remote Sens.*, vol. 6, no. 12, pp. 12055–12069, 2014, doi: 10.3390/rs61212055.
- [83] H. Lu and T. Koike, "Evaluation of AMSR2 and SMOS soil moisture products over Heihe River Basin in China," in *Proc. IEEE Int. Geosci. Remote Sens. Symp.*, 2015, pp. 685–688.
- [84] C. Ma, X. Li, L. Wei, and W. Wang, "Multi-scale validation of SMAP soil moisture products over cold and arid regions in Northwestern China using distributed ground observation data," *Remote Sens.*, vol. 9, no. 4, 2017, Art. no. 327, doi: 10.3390/rs9040327.
- [85] L. Zhang, C. He, M. Zhang, and Y. Zhu, "Evaluation of the SMOS and SMAP soil moisture products under different vegetation types against two sparse in situ networks over arid mountainous watersheds, Northwest China," *Sci. China Earth Sci.*, vol. 62, no. 4, pp. 703–718, 2019, doi: 10.1007/s11430-018-9308-9.
- [86] Y. Wang and G. Li, "Evaluation of simulated soil moisture from China land data assimilation system (CLDAS) land surface models," *Remote Sens. Lett.*, vol. 11, no. 12, pp. 1060–1069, 2020, doi: 10.1080/2150704X.2020.1820614.
- [87] Z. Wang, T. Che, T. Zhao, L. Dai, X. Li, and J.-P. Wigneron, "Evaluation of SMAP, SMOS, and AMSR2 soil moisture products based on distributed ground observation network in cold and arid regions of China," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 14, pp. 8955–8970, Aug. 2021, doi: 10.1109/JSTARS.2021.3108432.
- [88] B. D. Stocker, J. Zscheischler, T. F. Keenan, I. C. Prentice, S. I. Seneviratne, and J. Peñuelas, "Drought impacts on terrestrial primary production underestimated by satellite monitoring," *Nature Geosci.*, vol. 12, no. 4, pp. 264–270, 2019, doi: 10.1038/s41561-019-0318-6.

- [89] A. F. Feldman et al., "Moisture pulse-reserve in the soil-plant continuum observed across biomes," *Nature Plants*, vol. 4, no. 12, pp. 1026–1033, 2018, doi: 10.1038/s41477-018-0304-9.
- [90] R. M. Maxwell, L. E. Condon, and S. J. Kollet, "A high-resolution simulation of groundwater and surface water over most of the continental US with the integrated hydrologic model parflow v3," *Geoscientific Model Develop.*, vol. 8, no. 3, pp. 923–937, 2015, doi: 10.5194/gmd-8-923-2015.
- [91] Y. Chen, K. Yang, J. Qin, L. Zhao, W. Tang, and M. Han, "Evaluation of AMSR-E retrievals and GLDAS simulations against observations of a soil moisture network on the central Tibetan Plateau," *J. Geophysical Res.*: *Atmos.*, vol. 118, no. 10, pp. 4466–4475, 2013, doi: 10.1002/jgrd.50301.
- [92] W. A. Dorigo et al., "International soil moisture network: A data hosting facility for global in situ soil moisture measurements," *Hydrol. Earth Syst. Sci.*, vol. 15, no. 5, pp. 1675–1698, 2011, doi: 10.5194/hess-15-1675-2011.
- [93] J. Zeng, Z. Li, Q. Chen, H. Bi, J. Qiu, and P. Zou, "Evaluation of remotely sensed and reanalysis soil moisture products over the Tibetan Plateau using in-situ observations," *Remote Sens. Environ.*, vol. 163, pp. 91–110, 2015, doi: 10.1016/j.rse.2015.03.008.
- [94] S. V. Kumar, P. A. Dirmeyer, C. D. Peters-Lidard, R. Bindlish, and J. Bolten, "Information theoretic evaluation of satellite soil moisture retrievals," *Remote Sens. Environ.*, vol. 204, pp. 392–400, 2018, doi: 10.1016/j.rse.2017.10.016.
- [95] H. Cui et al., "Evaluation and analysis of AMSR-2, SMOS, and SMAP soil moisture products in the Genhe area of China," J. Geophysical Res.: Atmos., vol. 122, no. 16, pp. 8650–8666, 2017, doi: 10.1002/2017JD026800.
- [96] W. T. Crow et al., "Robust estimates of soil moisture and latent heat flux coupling strength obtained from triple collocation," *Geophysical Res. Lett.*, vol. 42, no. 20, pp. 8415–8423, 2015.
- [97] M. Kędzior and J. Zawadzki, "Comparative study of soil moisture estimations from SMOS satellite mission, GLDAS database, and cosmic-ray neutrons measurements at COSMOS station in Eastern Poland," *Geoderma*, vol. 283, pp. 21–31, 2016, doi: 10.1016/j.geoderma.2016.07.023.
- [98] X. Guo, D. Meng, X. Chen, and X. Li, "Validation and comparison of seven land surface evapotranspiration products in the Haihe River Basin, China," *Remote Sens.*, vol. 14, no. 17, 2022, Art. no. 4308.
- [99] R. M. Maxwell and S. J. Kollet, "Interdependence of groundwater dynamics and land-energy feedbacks under climate change," *Nature Geosci.*, vol. 1, no. 10, pp. 665–669, 2008.
- [100] E. Leonarduzzi et al., "Training machine learning with physicsbased simulations to predict 2D soil moisture fields in a changing climate," *Front. Water*, vol. 4, 2022, Art. no. 927113, doi: 10.3389/frwa.2022.927113.



processes modeling.

Yuan He received the B.S. degree in resources and environmental science from the Beijing University of Civil Engineering and Architecture, Beijing, China, in 2020. He is currently working toward the Ph.D. degree in natural resources with the Faculty of Geographical Science, Beijing Normal University, Beijing, China.

He is a Member of the State Key Laboratory of Earth Surface Processes and Resource Ecology, Beijing, China. His research interests include deep learning, remote sensing of lakes, and lake hydrodynamic



**Shuyan Peng** received the B.S. degree in geographic information science from Southwest University, Chongqing, China, in 2021. She is currently working toward the M.S. degree in natural resources with the Faculty of Geographical Science, Beijing Normal University, Beijing, China.

She is a Member of the State Key Laboratory of Earth Surface Processes and Resource Ecology, Beijing, China. Her current research interests include numerical simulation of groundwater-surface water interactions as well as soil moisture retrievals based

on microwave remote sensing.



Xiaofan Yang received the B.Eng. degree in automotive engineering from Tsinghua University, Beijing, China, in 2003, the M.S. degree in engineering mechanics from the Royal Institute of Technology, Stockholm, Sweden, in 2005, and the Ph.D. degree in mechanical engineering from Kansas State University, Manhattan, KS, USA, in 2010.

From 2010 to 2017, she was a Scientist with the U.S. Department of Energy (US DOE)'s Pacific Northwest National Laboratory, USA. She is currently a Professor in Hydrology with Beijing Normal

University, Beijing, China. She has authored or coauthored more than 70 papers in international peer-reviewed journals. Her research interests include subsurface hydrology, computational hydrology, and porous media, with specific focus on reactive transport modeling, groundwater modeling, flow and reactive transport in porous media, and multiscale modeling and simulations.

Dr. Yang is an Associate Editor for the *Journal of Hydrology: Regional Studies* and *Hydrological Processes*.



**Zheng Lu** received the M.S. degree in cartology and geographic information system in 2013 from Beijing Normal University, Beijing, China, where he is currently working toward the Ph.D. degree in natural resources with the Faculty of Geographical Science. He is a Member of the State Key Laboratory of

Earth Surface Processes and Resource Ecology, Beijing, China. His current research interests focus on the applications of integrated hydrologic models and remote sensing approaches to describe groundwatersurface water interactions and the associated pro-

cesses, especially changes in land surface processes, and assessing their possible impacts on the eco-hydrology over the Heihe River Basin of China as well as other arid and semiarid regions.